Author’s response to reviews

Title: "I can't bend it and it hurts like mad": Direct observation of gout consultations in routine primary health care

Authors:

Anthony Dowell (tony.dowell@otago.ac.nz)
Caroline Morris (caroline.morris@otago.ac.nz)
Lindsay Macdonald (lindsay.macdonald@otago.ac.nz)
Maria Stubbe (maria.stubbe@otago.ac.nz)

Version: 2 Date: 25 Jul 2017

Author’s response to reviews:

Uploaded as supplementary material